Author’s response to reviews

Title: Endoscopic balloon dilatation for benign hepaticojejunostomy anastomotic stricture using short double-balloon enteroscopy in patients with a prior Whipple's procedure: a retrospective study

Authors:
Sho Mizukawa (mimizukawa@gmail.com)
Koichiro Tsutsumi (ttmkuicru@yahoo.co.jp)
Hironari Kato (drkatocha@yahoo.co.jp)
Shinichiro Muro (shinm0416@yahoo.co.jp)
Yutaka Akimoto (qttsp974@gmail.com)
Daisuke Uchida (d.uchida0309@gmail.com)
Kazuyuki Matsumoto (matsumotokazuyuki0227@yahoo.co.jp)
Takeshi Tomoda (tomotake79@yahoo.co.jp)
Shigeru Horiguchi (horiguchis@gmail.com)
Hiroyuki Okada (hiro@md.okayama-u.ac.jp)

Version: 3 Date: 08 Jan 2018

Author’s response to reviews:
Jan 09, 2018

Tovah Aronin
Editor-in-Chief
BMC Gastroenterology

RE: Submission of revised Manuscript ID BMGE-D-17-00386
Dear Editor:

Thank you for your email dated 04-Jan-2018 in which you provided us with the reviewers’ comments.

We carefully reviewed the comments and revised the manuscript accordingly. Our responses are given in a point-by-point manner below.

We hope the revised manuscript is now suitable for publication in your esteemed journal and look forward to hearing from you.

Sincerely,

Dr. Koichiro Tsutsumi,
Department of Gastroenterology,
Okayama University Hospital,
2-5-1 Shikata-cho, Kita-ku,
Okayama-city, Okayama,
700-8558, Japan
Tel.: +81-86-235-7219
Fax: +81-86-225-5991
E-mail: ttmkuicru@yahoo.co.jp

Response to Reviewer: 1
Thank you for your helpful comments, which allowed us to significantly improve our manuscript.

Response to Reviewer: 3

Thank you for your helpful comments, which allowed us to significantly improve our manuscript.

Comment 1. On line 124, the blind and of the bowel described as ileum which instead should be jejunum.

Response: We thank the Reviewer for this insightful comment. In accordance with the comment, we made an alteration (Page9, Line124).

Response to Technical Comments:

Comment 1. Please, add the statement on consent to participate not only in the Methods section, but also in the Ethics approval and consent to participate section.

Response: We appreciate the comment on this point. In accordance with the comment, we added this information to the participate section (Page26, Line 357-358).

Comment 2. Please, confirm that the consent for publication obtained was written.

Response: We appreciate the comment on this point. This manuscript contains some individual person’s data (images). So we obtained consent for publication in writing form from the patient. In accordance with the comment, we added the consent form to the additional file.